Author's response to reviews

Title: The Efficacy of Motivational Counseling and SMS-reminders on Daily Sitting Time in Patients with Rheumatoid Arthritis: Study Protocol for a Randomized Controlled Trial

Authors:

- Bente Appel Esbensen (bente.appel.esbensen@regionh.dk)
- Tanja Thomsen (tanja.thomsen@regionh.dk)
- Merete L. Hetland (merete.hetland@dadlnet.dk)
- Nina Beyer (ninabeyer.privat@gmail.com)
- Julie Midtgaard (julie@ucsf.dk)
- Katrine Løppenthin (katrine.bjerre.loeppenthin@regionh.dk)
- Mikkel Østergaard (mo@dadlnet.dk)
- Poul Jennum (poul.joergen.jennum@regionh.dk)
- Jan Sørensen (jas@sdu.dk)
- Robin Christensen (robin.daniel.kjersgaard.christensen@regionh.dk)
- Mette Aadahl (mette.aadahl@regionh.dk)

Version: 5  Date: 19 December 2014

Author's response to reviews:

Thank you accepting our revised manuscript. We have made following formatting changes as suggested:

1. Our project has been approved by the Regional Committee on Biomedical Research Ethics (page 17).
2. We have funding information below the Acknowledgements section (page 21-22)

Please let me know if I can do anything further.

Kind regards

Bente Appel Esbensen